Author's response to reviews

Title: Evaluation of the utilization of the preanaesthetic clinics in a University Teaching Hospital

Authors:
- Seetharaman Hariharan (uwi.hariharan@gmail.com)
- Deryk Chen (uwideryk@yahoo.com)
- Lorna Merritt-Charles (lmerritt@tstt.net.tt)

Version: 4 Date: 3 May 2006

Author's response to reviews: see over
To
BioMed Central Editorial

Ref: MS: 3102722469315773 - Evaluation of the utilization of the preanaesthetic clinics in a University Teaching Hospital

Dear Madam/ Sir,
Many thanks for reviewing the above article and accepting it for publication. We have modified the article, and accordingly incorporated all the requirements of the journal in the present revised version.

1. We have formatted the article according to the “instructions to authors” including the title page and references
2. We have included the competing interests section, authors’ contribution as suggested

We sincerely hope that these changes will adequately satisfy the requirements of the journal.
We request that the present revised version may kindly be considered for publication.
Thanking you
Yours truly,

(S. Hariharan)